Author’s response to reviews

Title: Cardiac resynchronization therapy-defibrillator pocket infection caused by Mycobacterium fortuitum: a case report and review of the literature

Authors:

Zhu Jun (drzhujun@medmail.com.cn)
Qingluan Yang (10301010198@fudan.edu.cn)
Junjie Pan (junjie112@163.com)
Haiming Shi (13901824859@163.com)
Jin Bo (jinbo7711@aliyun.com)
Qi Ying Chen (chenqiying2006@163.com)

Version: 2 Date: 18 Dec 2018

Author’s response to reviews:

Dear editor:

Thank you for your suggestions.

We have revised the manuscript with all your comments.

First of all, we emphasize that this case is the first recorded case that re-implanted a CIED in the same side, which adds to the current knowledge on this subject. Secondly, We describe the searching method in detail. Thirdly, our manuscript has been revised by a native English speaker.

Thank you very much for your consideration.

Yours Sincerely,

Qiying Chen, M.D., Ph.D.
Department of Cardiology, Huashan Hospital of Fudan University
Shanghai 200040